Author's response to reviews

Title: Effect of commercial breakfast fibre cereals compared with corn flakes on postprandial blood glucose, gastric emptying and satiety in healthy subjects: a randomized blinded crossover trial

Authors:

Joanna Hlebowicz (joanna.hlebowicz@skane.se)
Jennie Wickenberg (jennie.wickenberg.647@student.lu.se)
Rickard Fahlstrom (lak02rfa@student.lu.se)
Ola Bjorgell (Ola.Bjorgell@med.lu.se)
Lars-Olof Almer (Lars-Olof.Almer@med.lu.se)
Gassan Darwiche (Gassan.Darwiche@skane.se)

Version: 5 Date: 6 September 2007

Author's response to reviews:

Dear BioMed Central Editorial Production Team 2007.09.06 Malmö

I would like to thank you for your important and valuable comments regarding the content of the present manuscript. We have clearly been able to reply to all your remarks, as follows below.

Please notice that the changes in the manuscript have been done as required.
1. Different email address for each author are given.
2. Figures is changed to Figure legends.

Sincerely,
Joanna Hlebowicz, MD
Dept. of Medicine, Malmö University Hospital
Ingång 35, 205 02 Malmö, Sweden
E-mail: Joanna.Hlebowicz@med.lu.se